Author's response to reviews

Title: Protocol: Transmission and prevention of influenza in Hutterites: Zoonotic transmission of influenza A: swine & swine workers

Authors:

Margaret L Russell (mlrussel@ucalgary.ca)
Julia Keenliside (julia.keenliside@gov.ab.ca)
Richard Webby (Richard.webby@stiude.org)
Kevin Fonseca (k.fonseca@provlab.ab.ca)
Pam Singh (singhp@mcmaster.ca)
Lorraine Moss (mossl@mcmaster.ca)
Mark Loeb (loebm@mcmaster.ca)

Version: 4 Date: 11 November 2009

Author's response to reviews:

RE: Formatting Changes for MS: 1160003449310726

1. REQUESTED CHANGE: Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

RESPONSE: These changes have been made as requested.

2. REQUESTED CHANGE: Qualifications: Please remove the authors’ qualifications (e.g. PhD / MD) from the manuscript and submissions system.

RESPONSE: These changes have been made as requested.

3. REQUESTED CHANGE: Keywords: Please remove these.

RESPONSE: Keywords have been removed.

4. REQUESTED CHANGE: Abbreviations: Please format these in to a paragraph.

RESPONSE: The listing of abbreviations has been removed. All abbreviations are introduced in the body of the text.

5. REQUESTED CHANGE Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

RESPONSE: The manuscript has been checked and some corrections made.

6. Although not a requested change we note that the manuscript as submitted included a statement of competing interests on both the title page and in a specific section of the manuscript, preceding Author’s Contributions. The statement of competing interests has been removed from the title page.